Author's response to reviews

Title: Young men's attitudes and behaviour in relation to mental health and technology: implications for the development of online mental health services

Authors:

Louise A Ellis Dr (louise.ellis@sydney.edu.au)
Philippa Collin Dr (P.Collin@uws.edu.au)
Patrick J Hurley Mr (patrick.hurley@sydney.edu.au)
Tracey A Davenport Ms (tracey.davenport@sydney.edu.au)
Jane M Burns Ass Prof (Jane@yawcrc.org.au)
Ian B Hickie Prof (ian.hickie@sydney.edu.au)

Version: 3 Date: 21 March 2013

Author's response to reviews: see over
21 March, 2013

Re MS: 4398953727869685

Please find attached a resubmission of the manuscript “Young men’s attitudes and behaviour in relation to mental health and technology: implications for the development of online mental health services”.

We thank you for the opportunity to submit a revised version of our article. Please find below our responses to Harald Baumeister’s comments. Note the other reviewers were happy with the changes made.

Comment 1: There is still some information missing concerning the response options and scoring frames used for the following survey sections: “Interests and technology use” and “Technology and mental health”.
Response: We have added more detail regarding the response options and scoring frames for the two survey sections.

Comment 2: In the abstract you mentioned 18 focus groups whereas in the methods, results and discussion sections and table 1 you wrote that there were 17 groups.
Response: There were 17 focus groups so we have changed this in the abstract

We thank the reviewers for their helpful comments throughout this process and look forward to hearing from you soon.

Yours sincerely,

Dr Louise Ellis